Author's response to reviews

Title: Periostin is up-regulated in high grade and high stage prostate cancer

Authors:

Verena Tischler (verena.tischler@usz.ch)
Florian R Fritzsche (florian.fritzsche@usz.ch)
Peter J Wild (peter.wild@cell.biol.ethz.ch)
Carsten Stefan (cstephan6@yahoo.com)
Hans-Helge Seifert (hans-helge.seifert@hbh-kliniken.de)
Marc-Oliver Riener (Marc-Oliver.Riener@uk-erlangen.de)
Thomas Hermanns (thomas.hermanns@usz.ch)
Ashkan Mortezavi (ashkan.mortezavi@usz.ch)
Josefine Gerhardt (josefine.gerhardt@usz.ch)
Peter Schraml (peter.schraml@usz.ch)
Klaus Jung (klaus.jung@charite.de)
Holger Moch (holger.moch@usz.ch)
Alex Soltermann (alex.soltermann@usz.ch)
Glen Kristiansen (glen.kristiansen@usz.ch)

Version: 6 Date: 2 June 2010

Author's response to reviews: see over
Dear Editorial Production Team,

Thank you very much for reviewing our revised manuscript. Please find details to the manuscript changes below:

Major revisions (we require the author to make these changes)

Tables: table 1 has been divided into parts, a and b. It is important that tables are numbered in ascending numerical order: 1, 2 and 3. Please renumber the tables and any references to them in the text. Alternatively, please merge the tables into one larger table.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Please make sure you use the latest version of the main file as we have made several formatting changes and using a previous unformatted version will delay publication.

The table numbers were changed according to your requirements. All changes are marked in red colour. In addition, two sentences for better understanding of the results part were included (marked red).

We hope that we fulfilled all necessary changes. Please let us know if we can further support the editing process.
Yours sincerely,

Verena Tischler, MD

Glen Kristiansen, MD